Author's response to reviews

Title: Associations between perceived and observational physical environmental factors and the use of walking paths A cross-sectional study

Authors:

Yingnan Jia (12111020010@fudan.edu.cn)
Hua Fu (hfu@fudan.edu.cn)

Version: 2
Date: 6 January 2014

Author's response to reviews:

Dear Rita Aguirre:

Thank you very much for your attention and comments on our paper. We have revised the manuscript according to your kind advices.

A statement to ethical permission has been added in the section of method. Competing interests and authors' contributions have also been revised.

We sincerely hope this manuscript will be finally acceptable to be published on BMC Public Health. Thank you very much for all your help and looking forward to hearing from you soon.

Best regards
Sincerely yours
Dr. Yingnan Jia